

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Romagnoli, Matteo

Working Paper Can electricity liberalisation foster the development of radical clean-energy technologies?

Working Paper, No. 044.2022

Provided in Cooperation with: Fondazione Eni Enrico Mattei (FEEM)

Suggested Citation: Romagnoli, Matteo (2022) : Can electricity liberalisation foster the development of radical clean-energy technologies?, Working Paper, No. 044.2022, Fondazione Eni Enrico Mattei (FEEM), Milano

This Version is available at: https://hdl.handle.net/10419/267509

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU





Working Paper 044.2022

Can electricity liberalisation foster the development of radical clean-energy technologies?

Matteo Romagnoli

Can electricity liberalisation foster the development of radical clean-energy technologies?

By **Matteo Romagnoli**, Department of Economics, Management and Statistics, Università degli Studi di Milano-Bicocca

Summary

The paper investigates the effect of electricity liberalisation on the variety of clean energy patent' search space to asses whether a more competitive electricity market can foster the development of radical clean-energy technologies. This idea is tested using a cross-section of patents filed in the period 1990-2017, a set of patent-level indicators and an instrumental variable approach. Results show that electricity liberalisation pushes clean-energy patents to cite knowledge from technological fields other than their own. However, the reform does not significantly affect the overall breath of the knowledge base of these patents. Additional insights are drawn by looking at the correlation between electricity liberalisation and an indicator of novelty in patents' search space. The results are consistent with the claim that electricity liberalisation has a positive effect on the development of radical clean-energy technologies. At the same time, by describing how the reform changes clean-energy patents' search space, they define this effect more precisely.

Keywords: Clean-energy Technologies, Electricity Liberalisation, Climate Change, Patent Data

JEL Classification: L94, 031, Q42, Q55

Address for correspondence: Matteo Romagnoli PhD Candidate, Department of Economics, Management and Statistics Università degli Studi di Milano-Bicocca Piazza dell'Ateneo Nuovo 1, 20126 Milan (Italy) e-mail: <u>m.romagnoli4@campus.unimib.it</u>

The opinions expressed in this paper do not necessarily reflect the position of Fondazione Eni Enrico Mattei

Can electricity liberalisation foster the development of radical clean-energy technologies? *

Matteo Romagnoli[†]

December 18, 2022

Abstract

The paper investigates the effect of electricity liberalisation on the variety of cleanenergy patent' search space to asses whether a more competitive electricity market can foster the development of radical clean-energy technologies. This idea is tested using a cross-section of patents filed in the period 1990-2017, a set of patent-level indicators and an instrumental variable approach. Results show that electricity liberalisation pushes clean-energy patents to cite knowledge from technological fields other than their own. However, the reform does not significantly affect the overall breath of the knowledge base of these patents. Additional insights are drawn by looking at the correlation between electricity liberalisation and an indicator of novelty in patents' search space. The results are consistent with the claim that electricity liberalisation has a positive effect on the development of radical clean-energy technologies. At the same time, by describing how the reform changes clean-energy patents' search space, they define this effect more precisely.

Keywords: Clean-energy Technologies, Electricity Liberalisation, Climate Change, Patent Data. **Jel Codes**: L94, 031, Q42, Q55

*I would like to thank Antoine Dechezleprêtre and Tobias Kruse for allowing access to latest version of the OECD Environmental Policy Stringency database. I thank Dennis Verhoeven for the data and the help in replicating the patent indicators presented in Verhoeven et al (2016). I acknowledge the very useful suggestions from Matteo Manera, Anna Cretì, Marc Baudry, Francesco Nicolli, Nicoló Barbieri and Ugo Rizzo as well as of the partecipants at the College of Europe and RNI, 2022 Forum Innovation. Part of this research was conducted while I was a visiting Phd Student at the Climate Economics Chair (Institut Louis Bachelier, Paris, France).

[†]Dept. of Economics, Management and Statistics, Universita' degli Studi di Milano-Bicocca, Piazza dell'Ateneo Nuovo 1, 20126 Milan (Italy). Email address m.romagnoli4@campus.unimib.it

1 Introduction

The optimal regulatory framework for the electricity market is an hot topic of discussion among academics, policy makers and market agents. In the European Union the recent power price crisis, triggered by the Russian invasion of Ukraine in February 2022, has reignited the discussion on whether the current design of the electricity market is fitfor-purpose (ACER, 2022). ¹ In 2021, the design of the electricity market has been a topic of discussion also in the U.S., when the Texas blackout sparked a debate on the role played by electricity liberalisation in the failure of the power system.

There are many dimensions according to which one can judge the performance of an electricity market; among others: security of supply, affordability, ability to attract investments, number and type of services offered, quantity and quality of innovation. By shedding light on the relationship between different regulatory frameworks and each of these dimensions, the academic literature has brought and can bring relevant contributions to the aforementioned debate. This is the aim of this paper, which studies the relationship between electricity regulation and innovation, focusing in particular on the effect of electricity liberalisation on the development of radical clean-energy technologies.

The importance of having a regulatory framework in the electricity sector able to foster the development of radical clean-energy innovations can not be overstated. Radical technologies are widely recognized to be fundamental in triggering technological and societal changes (Acemoglu et al 2020, Rizzo et al., 2018, Arthur 2007, Olsson 2000, Nelson and Winter 1982). As way of example, the European Union Innovation Fund, one of the world's largest funding programs aimed at supporting low-carbon innovations, explicitly focuses on "highly innovative technologies" and "breakthrough technologies" ². Breakthrough clean-energy innovations are a major tool at our disposal to decarbonise electricity supply, an objective of the utmost importance given the high contribution of electricity generation to global greenhouse gas emissions ³.

Part of the complexity in the development of highly innovative clean-energy technologies comes from the fact that the electricity sector is a large technical system, i.e. a system characterized by capital-intensive infrastructures, a variety of actors and many

¹As an example, the last "Assessment of the EU wholesale Electricity Market Design" by ACER (Agency for the Cooperation of Energy Regulators") is focused on precisely this issue (ACER, 2022).

²See:https://www.euinnovationfund.eu/ and https://ec.europa.eu/commission/ presscorner/detail/en/ip_22_4402

³https://ourworldindata.org/emissions-by-sector;https://www.iea.org/reports/ global-energy-review-co2-emissions-in-2021-2

complex technical components all of which interact together (Markard and Truffer, 2006; Joerges, 1998; Hughes, 1987). Due to these features, large technical systems tend to suffer from inertia and path-dependence, with technological progress that is often characterized by incremental improvements (Hughes, 1987).

In this context, it has been hypothesised that electricity liberalisation, by increasing the variety of clean-energy technologies' search space, can act as a driver for the development of radical innovations in this field (Negro et al., 2012; Markard and Truffer, 2006). The search space of a patent can be defined as the totality of knowledge inputs on which the invention relies, i.e. the set of the sources used to developed the invention. A diverse search space is characterised by knowledge inputs coming from different technological fields and it is therefore evidence that the patent is building on different strand of knowledge. A well-established body of literature in innovation theory highlights how some characteristics of the search space of an invention, such as breadth and complexity, are linked with its radicalness, originality and novelty (Barbieri et al., 2020, Verhoeven et al 2016 Squicciarini et al., 2013 Shane, 2001, Trajtenberg et al.,1997).

In a nutshell, the main idea behind the effect of interest can be expressed as follows. In regulated markets, monopolists have low incentives to develop radical clean-energy technologies that are far from their knowledge base and can jeopardize their exiting assets. Conversely, they are more likely to deal with the problem of decarbonizing energy supply with incremental innovations that are more compatible with their existing asset base and therefore less costly for them to develop and adopt (Nesta et al., 2014; Negro et al., 2012). Such incremental improvements are generally the results of patents with a narrow search space, because they strongly rely on the knowledge already available in their technological fields (Squicciarini et al 2013; Shane 2001). The liberalisation of the electricity market, by allowing new players to enter into the market and by creating a more competitive environment, can favor a broader approach to R&D and a wider search space in the development of clean-energy technologies, thus leading to more radical clean-energy innovations (Negro et al., 2012; Markard and Truffer, 2006). Section 2 discusses this point more in depth.

This paper contributes to the literature on electricity liberalisation and innovation by being the first one to test this idea empirically, exploiting patent data and a wellestablished set of patent-level indicators (see Verhoeven et al., 2016; Squicciarini et al., 2013). This kind of indicators have been extensively used in the innovation literature (see for instance Barbieri et al., 2020; Verhoeven et al., 2016; Squicciarini et al., 2013; Harhoff e tal 2003; Hall at al., 2001; Shane, 2001; Trajtenberg et al., 1997) and allow me to study how the search space of clean-energy technologies changes as the electricity market becomes more competitive.

To uncover the casual effect of interest, I rely on an Instrumental Variable (IV) approach. The proposed strategy follows Nicolli and Vona (2019) and uses regulation in telecommunication as an instrument for regulation in electricity. The intuition behind this approach is that the reform of the telecommunication sector occurred before the one of the electricity market and was instrumental in giving momentum to the latter (Pollitt 2012, Joskow 2008). At the same time, this reform can be considered independent from the direct lobbying power of actors in the energy sector (Nicolli and Vona 2019) and from technological developments in energy-supply technologies. As robustness check, I also use regulation in air transport as instrument and provide the results in Appendix A.5.

Results from the empirical analysis are consistent with the claim that electricity liberalisation can foster radical clean-energy innovations. In particular, they show that the reform pushes clean-energy patents to cite more knowledge from outside technological fields, i.e. technological fields other than the ones they are allocated to; a pattern that has been found to lead to more radical inventions (Squicciarini et al., 2013; Shane, 2001). In addition, by describing in depth the effect of the reform on the search space of these patents, the results also point out the limits of this effect.

The remainder of the paper is organized as follows: Section 2 reviews the relevant literature and formalizes the research hypothesis. Section 3 describes the data used and Section 4 discusses the empirical strategy. Section 5 is dedicated to the IV strategy, Section 6 presents the results and Section 7 concludes.

2 Liberalisation and Innovation in the Electricity Sector

The electricity sector is one of the best examples of a large technical system (Hughes, 1987). Joerges (1998) defines large technical systems as complex systems of physical structure and machinery, integrated over space and time and supporting other technical systems. The strong interactions between the different components of a large technical system, as well as the interactions between the system itself and the other technical systems it supports, make innovation in these environments often characterised by incremental improvements rather than radical changes (Negro et al 2012, Markard and Truffer 2006, Hughes 1987).

That being said, there are many examples of radical innovations being developed in large-technical systems. Hughes (1987) identifies "reverse salients" as the main sources of these radical innovations ⁴. Markard and Truffer (2006) focus instead on external factors that can act as triggers for this kind of inventions. Some example of these external factors include technological developments in related fields, government intervention or changes in the preferences of the customer base. In light of this discussion, Markard and Truffer (2006) look at the electricity sector and analyze a series of 44 interviews collected in more than 30 utilities ⁵, presenting case study evidence focused on fuel cell innovation strategy. They conclude that the liberalisation of the electricity sector can be one of these external driver which increases the variety of the search space of clean-energy technologies, thus fostering the development of more radical clean-energy innovations. A similar argument is presented by Negro et al. (2012). In this regard, the positive relationship between the variety of the search space of an invention and its radicalness has been widely documented and it is now a well-known phenomenon (Barbieri et al., Verhoeven et al., 2016, 2020, Squicciarini et al., 2013, Shane, 2001; Trajtenberg et al., 1997). Section 3.1 discusses this point more in depth.

A first channel through which electricity liberalisation can impact the search space of clean-energy patents is by allowing new entrants in the market. The importance of new

⁴Reverse salient are defined as "those components that lagged behind other components in an expanding system, thus threatening the possibility of expansion for the whole system" (Hughes, 1987 - Summary). As an example Hughes (1987) brings the early days of electricity supply "when the prevailing direct current (DC) technology was not able to transmit ever more growing energy flows efficiently over long distances" (Hughes, 1987 - Summary).

⁵The utilities are from Germany, the Netherlands and Switzerland

entrants for the development of radical innovations is largely acknowledged in the literature (e.g. Acemouglu et al., 2022; Akcigit and Kerr, 2018; Klepper 1996; Winter, 1984). The model presented by Klepper (1996) in particular predicts that, when competition is low, the diversity of R&D will be compromised. A narrow approach to R&D is likely to be particularly detrimental for clean-energy technologies because their development requires a variety of knowledge inputs and competences that are far from the traditional knowledge base of incumbents (Barbieri et al., 2020; De Marchi, 2012). An increase in competition can instead be expected to have the opposite effect. New entrants have a comparative advantage in the adoption of radical technologies and consequently they can foster their development (Acemouglu et al., 2022; Akcigit and Kerr, 2018). More generally, a key determinant of radical innovations at the firm level appears to be how "open to disruption" the firm is, and new entrants often have more to gain by disrupting the status-quo (Acemouglu et al., 2022). This can be expected to lead to a wider approach to R&D and a wider search space, especially in the case of the electricity sector, where new entrants are not tied to the traditional large-scale plants and technologies generally used by incumbents (Nesta et al., 2014; Nicolli and Vona, 2016). The literature on the relationship between innovation and product cannibalisation is also relevant to this point as radical clean-energy technologies are often competence-destroying for incumbents in the electricity market (Nesta et al., 2014). When cannibalisation is an issue, competitive pressure has been shown to be essential in order to push incumbents to innovate (Conner 1988, Reinganum, 1983), which suggests that more competition in the electricity market can also have a positive impact on the search space of clean-energy patents developed by electric utilities.

The entry of new players is not the only channel through which liberalisation can foster the development of more radical clean-energy technologies. Dolphin and Pollitt (2020) use patent data from the UK and show that, after the reform of the electricity market, innovation activity shifted from regulated monopolist to electric equipment manufacturer. The latter are less tied to traditional generation technologies and have different incentives with respect to the former, thus they can be expected to rely on a wider search space.

Finally, privatisation can change the innovation environment within the incumbents and make them more likely to rely on a wider knowledge base. This is in line with the evidence suggesting that investor-owned electric utilities are more responsive to renewable policies than state-owned utilities (Nicolini and Tavoni 2017, Delmas and Montes-Sancho al 2011, Carley 2009). Existing empirical studies on the relationship between electricity liberalisation and innovation have mainly looked at the effect of the reform on the "quantity" of innovations developed (e.g. number of patents) or the amount of inputs used in the research process (e.g. investment in R&D). A useful way to categorize this literature is by distinguish between studies focused on clean-energy technologies and studies that look at all kinds of innovations developed in the electricity sector.

Among the latter, a first body of literature finds a decrease in R&D expenditures and overall patent activity after the electricity market has been liberalised, e.g., Sanyal and Gosh (2013), Sterlacchini (2012), Jamasb and Pollitt (2008) and Dooley (1998). ⁶ A recent paper by Marino et al (2019) highlights the presence of an inverted-U relationship between electricity liberalisation and the number of patents developed in the electricity sector. Wang and Mogi (2017) find instead an increase in patenting from Japanese electric utilities after liberalisation coupled however with a decrease in R&D expenditures.

The literature focused on the relationship between liberalisation and clean-energy technologies shows instead a positive effect of market structure reforms on patenting. The heterogeneous effect of electricity liberalisation on clean-energy technologies and traditional technologies is now a well established fact in the literature (see for instance Li et al 2020). Nicolli and Vona (2016) find that lowering entry barriers has a positive effect on patents in renewable energy technologies and this effect is stronger for those technology that are characterised by the potential entry of small and independent power producers. Nesta et al (2014) show that liberalisation in the electricity sector led to an higher number of patents in clean-energy technologies and, when paired with environmental policies, it also increased the number of clean-energy triadic-patent families. A similar result is highlighted by Jamasb and Pollitt (2011) in the context of the UK electricity reform. Finally, Jacobsson and Bergek (2004) provide anecdotal evidence of the role played by new entrants in the evolution of the German wind energy sector.

Based on this discussion, the research hypothesis of the paper is that electricity liberalisation widens the search space of clean-energy patents, thus being a driver for the development of radical clean-energy technologies. Section 3.1 discusses the patent indicators used to measure the breadth and complexity of a patent's search space and

⁶Among these, the paper by Sanyal and Gosh (2013) is the only one that also measures the effect of liberalisation on patent quality using patent-level indicators. Focusing on the US they find a negative effect of the 1992 Energy Policy Act on the average number of forward citations and the generality of patents filed by electric equipment manufacturer.

how these characteristics relate to the radicalness, originality and novelty of an invention. The same section also discusses more in depth how these indicators are built and their interpretation.

3 Data and Descriptive Statistics

Using the database PATSTAT, I gathered data on clean-energy patent applications filed at the European Patent Office (EPO) between 1990 and 2017. I follow a common approach in the literature and identify clean-energy technologies through the code "Y02E" of the Cooperative Patent Classification (see for instance Calel and Dechelepretre 2016, Dechezleprêtre et al 2017). The resulting cross-section of patents was merged with the OECD Patent Quality Indicators database (February 2022) (Squicciarini et al., 2013), which contains information on a variety of patent quality indicators computed at the application level. Finally, information on the patent indicators developed by Verhoeven et al. (2016) were made available by the authors and added to the dataset. Section 3.1 discusses in detail each indicator used in the analysis.

To avoid double counting of the same invention, only one application was selected from each patent family. ⁷ A well-known issue with this approach is that patent applications belonging to the same family often display different values of the same indicator. Following Barbieri et al. (2020) and Verhoeven et al. (2016) I select within each family only the application with the highest value of the indicator of interest. Section A.5 of the Appendix shows that the results are robust also to selecting the application with the lowest value of the indicator of interest.

Figure 1 plots the number of clean-energy patents in the sample by year and Y02E 6digits sub-class. As one would expect, the classes Y02E/10 (Energy generation through renewable energy sources) and Y02E/60 (Enabling technologies; Technologies with a potential or indirect contribution to GHG emissions mitigation) are by far the ones with more patent families filed in the period of interest.⁸

Each patent is assigned to a country based on the address of the inventor(s). Using the country of the inventor(s) rather than the one of the applicant connects the document with the environment in which it was developed and allows to make a direct link with the

⁷A patent family is defined by the EPO as "a collection of patent applications covering the same or similar technical content", see https://www.epo.org/searching-for-patents/helpful-resources/first-time-here/patent-families.html

 $^{^{8}}$ In this regard, note that the class Y02E/60 contains all patents regarding energy storage.

notion of territory (Baudry and Dumont, 2006). Table 1 breaks down the patent families in the sample by inventor's country and shows that more than 60% of the families in the sample come from Germany and Japan combined, two well-known leaders in the development of clean-energy technologies. ⁹. When assigning a patent to the country of origin of the inventor(s), shared applications with inventors in different countries pose a problem. Since they are a relatively rare occurrence in the dataset, the analysis that follows is focused only on patents that I was able to unambiguously assigned to a single country. In Section A.6 of the Appendix I also consider patents with inventors in different countries and show that doing so does not change the results of the analysis.

The name of the applicant(s) for each patent application has been retrieved using the OECD HAN database (February 2022). ¹⁰ Table 2 breaks down the total number of applicants (3688) according to how many patent families they have filed in the period of interest and Table 3 provides the list of the top ten applicants in the sample in terms of patent families filed. Having information on the applicant allows me to build the cumulative stock of clean-energy patent families at the applicant level. In order to do this I apply the perpetual inventory method using a depreciation rate of 15%. ¹¹ This value for the depreciation rate is commonly used in the literature; see for instance Kafouros et al (2021), Hussinger and Pacher (2019), Hall (2005).

Finally, the resulting dataset was enriched with a variety of country level control variables. Among these we have the PMR index for the electricity sector (henceforth PMR_{elec}) i.e., the main independent variable of interest of this study. The OECD PMR database (2018) contains an array of time-varying sector-specific indicators computed to measure the level of liberalisation in different sectors of the economy (Vitale et al., OECD, 2020). The use of the PMR_{elec} index is very common in the literature studying electricity liberalisation, see for instance Marino et al., 2019, Nicolli and Vona 2019, Nicolli and Vona 2016, Nesta et al, 2014. In its last iteration, the indicator covers the period 1975-2018 and ranges from 6 to 0, with higher levels signalling a more regulated

⁹Patents from the United States and South Korea are not included in the sample because, at the time of writing, the index of the OECD Product Market Regulation database (Vitale et al., OECD, 2020) measuring liberalisation in the electricity sector is not computed for these countries.

¹⁰For applications with more than one applicant we focus on the applicant that filed the highest number of patent families in the period of interest. Note that more than 92% of the applications in our sample are linked to only one applicant and more than 99% are linked to at most two applicants.

¹¹According to the perpetual inventory method the patent family stock for firm *i* at time *t* (K_{it}) can be computed as $K_{it} = P_{it} + (1-\delta)K_{i,t1}$, where P_{it} is the number of patent families developed by firm *i* in year *t* and δ is the depreciation rate.

electricity market. Figure A.1 in the Appendix plots the evolution of the PMR_{elec} index by country in the period of interest.

Descriptive statistics for the variables discussed in this section, together with other variables used in the subsequent analysis are presented in Table 4.



Figure 1: Number of patent families by year and 6-digits Y02E code

Notes: Each patent application is allocated to all the Y02E 6-digits technological fields listed in the document.

Country	N°of Patent Families	Share
Australia	525	0.94
Austria	721	1.30
Belgium	510	0.92
Canada	1117	2.01
Denmark	2435	4.38
Finland	588	1.06
France	4798	8.62
Germany	13916	25.01
Ireland	115	0.21
Italy	1616	2.90
Japan	21364	38.40
Netherlands	1312	2.36
Norway	357	0.64
Poland	95	0.17
Spain	1103	1.98
Sweden	1095	1.96
Switzerland	1143	2.05
United Kingdom	2832	5.09
Total	55642	100.00

Table 1: Patent Families by Country

_

_

Notes: The table breaks down the overall number of patent families in the sample according to the inventor(s)' country.

N° Patent Families	N [°] of Applicants	Total Number of Families Filed
1	6,059	6,059
between 2-5	2604	7,181
between 6-10	477	3,592
between 11-15	178	2,289
between 16-20	90	1,592
between 21-25	62	1,413
between 26-30	45	1,252
between 31-35	38	1,271
between 36-40	23	879
between 41-45	11	468
between 46-50	17	820
between 51-55	11	579
between 56-60	11	633
between 61-65	10	633
between 66-70	13	880
between 71-75	5	367
between 76-100	27	2,351
between 101-200	33	4,841
between 201-300	13	3,285
between 301-500	8	2,932
between 501-1000	7	4,659
> 1000	5	7,666
Total	3688	55,642

Table 2: Breakdown of applicants by number of clean-energy patent families filed.

Notes: The table breaks down the number of applicants in the sample according to the number of clean-energy patent families filed between 1990 and 2017. Column 3 reports the overall number of patent families filed by applicants in each category.

Table 3: Top 10 Applicants for n° of Y02E patent families filed (1990-2017)

Name	Country	N° of Patent Families
1) Siemens AG	Germany	2022
2) Matsushita Elect	Japan	1996
3) CEA	France	1271
4) Toyota Jidosha	Japan	1252
5) Nissan Motor	Japan	1125
6) Vestas Wind Systems	Denmark	805
7) Mitsubishi Heavy Ind	Japan	774
8) Toshiba	Japan	767
9) Robert Bosch GMBH	Germany	688
10) Sony Copr	Japan	594

Notes: The table shows the name, country and number of patent families filed by the top 10 applicants in the sample for number of of Y02E patent families filed between 1990 and 2017.

Variable	Variable Description	Observations	Mean	Std. Deviation	Min	Max	Source
Radicalness	Radicalness Index (Squicciarini 2013, Shane 2001)	55,642	0.32	0.25	0	1	OECD Patent Quality database (Feb. 2022)
Originality	Originality Index (Squicciarini 2013, Trajtenberg et al.,1997)	$55,\!634$	0.69	0.21	0	0.98	OECD Patent Quality database
PMR _{elec}	Regulation in the Electricity Sector	$55,\!642$	1.36	0.96	0.14	6.28	OECD PMR database (2018)
EPS_{tech}	Stringency of policies supporting green technologies	$55,\!642$	3.16	1.31	0.5	6	OECD EPS database (2022)
GDP_{pc}	GDP per capita PPP (constant 2017 international \$)	55,642	43,300.38	6,350.86	23,064.95	7,7749.2	World Bank
Oil price (imports)	Crude oil import prices (\$/barrel)	55,642	67.28	32.70	11.8	117.78	OECD Data
Patent Scope	N° of IPC 4-digit codes to which a patent is assigned	$55,\!642$	2.01	1.23	1	15	OECD Patent Quality database
Bwd Citations	N° of citations to older patents	$55,\!642$	6.95	8.04	0	485	OECD Patent Quality database
N° of Applicants	N° of applicants listed in the patent	$55,\!642$	1.08	0.34	1	13	OECD Han database (Feb. 2022)
Patent family stock	Cumulative count of applicant's Y02E families	$55,\!642$	80.49	139.27	1	785.44	Author's calculations
Novelty Technological Origins	Verhoeven et al. (2016) NTO indicator	45,574	0.29	0.45	0	1	Verhoeven et al. (2016)

 Table 4: Descriptive Statistics

3.1 Examining the search space of clean-energy technologies

To measure the effect of liberalisation on the search space of clean-energy patents I rely on two well-established indicators computed in the OECD Patent Quality Indicators database (February 2022) (Squicciarini et al., 2013). These are the Radicalness Index (a là Shane 2001) and the Originality Index (a là Trajtenberg et al.,1997). In addition, to further investigate the main results obtained using these two indexes, I exploit the Novelty in Technological Origins (NTO) indicator developed by Verhoeven et al., (2016).

Note that the research question of the paper naturally leads to the use of indicators that rely on ex-ante characteristics of an invention to define its radicalness and novelty. This ex-ante approach defines radical innovations in terms of the characteristics of the underlying knowledge recombination process and therefore it is directly concerned with the search space of the invention (Verhoeven et al,. (2016). For a detailed discussion on the ex-ante and ex-post approaches to define radicalness see Barbieri et al. (2020) and Verhoeven et al,. (2016).

The remainder of this section discusses the three indicators that will be used as dependent variables in the study.

3.1.1 Radicalness Index

The radicalness index measures the radicalness of a patent looking at how how much it differs with respect to the patents it cites (Shane 2001). The intuition behind this indicator is that "when a patent cites previous patents in classes other than the ones it is in, that pattern suggests that the invention builds upon different technical paradigms from the one in which it is applied" (Shane, 2001, p. 210. See also Barbieri et al., 2020, Verhoeven et al, 2016, Squicciarini et al. 2013, Rosenkopf and Nerkar 2001).

Following the definition of Shane (2001), the index for a focal patent p, with J representing the set of patents cited by patent p and j = 1, 2, 3...J, is defined by Squicciarini et al. (2013) as:

$$Radicalness_p = \sum_{i}^{n_p} CT_j / n_p; \quad IPC_{pj} \neq IPC_p \tag{1}$$

Where CT_j is the count of IPC-4 digit codes (IPC_{*pj*}) of patent *j* that are not allocated to the focal patent *p*, weighted by the times each IPC-4 digit code appears at the most disaggregated level available in the backward citations of patent *j*. The denominator, n_p , is the count of the overall IPC classes in the backward citations of patents belonging to the set *J*, counted at the most disaggregated level available. The indicator is therefore normalised so that its value ranges from zero to one.

High levels of this index signify that the patent takes knowledge from outside technological fields and applies it to its own technological fields.

Based on the research hypothesis, electricity liberalisation is expected to widen the search space of clean-energy patents. Part of this process could entail also the exploration of outside technological fields, thus we expect the reform to have a positive effect on the radicalness index.

3.1.2 Originality Index

The originality index measures how much the backward citations of a patent are spread across different technological fields (Trajtenberg et al.,1997). The intuition behind this indicator is that knowledge recombination processes relying on a diversified set of knowledge sources are supposed to lead to more original outcomes (Barbieri et al., 2020, Dechezleprêtre et al., 2017; Verhoeven et al., 2016, Squicciarini et al. 2013, Trajtenberg et al.,1997).

Building on Hall at al. (2001), Squicciarini et al. (2013) compute the originality indicator as follows:

$$Originality_p = 1 - \Sigma_j^{n_p} s_{pj}^2 \tag{2}$$

Where s_{pj} is the percentage of citations made by patent p to patent class j out of the n_p IPC 8-digit patent codes contained in the patents cited by patent p. Note that the indicator is built starting from an Hirschman-Herfindahl Index that measures the extent to which the backward citations of the patent are concentrated among different technological fields (i.e., $\Sigma_j^{n_p} s_{pj}^2$). This being the case, the indicator ranges from zero to one and higher values of the indicator signal patents with backward citations spread across many different fields.

Based on the research hypothesis, we expect liberalisation to increase the variety of knowledge inputs that go into clean-energy technologies. This could push these technologies to cite a wider array of technological fields, which would positively impact the originality index.

3.1.3 Novelty in Technological Origins

The NTO indicator was developed by Verhoeven et al. (2016) in order to identify novel patterns of citations in patents, i.e., pattern of citations that have never occurred before. The emergence of such patterns suggests that the underlying technology uses a new or different approach than than those used by its predecessors (Rizzo et al. 2018, Verhoeven et al. 2016, Arthur 2007).

With respect to the radicalness index, Verhoeven et al. (2016) argue that "citing from 'outside' fields of knowledge is not a sufficient condition to actually apply a novel approach since a large number of previous patents might have already sourced knowledge from these 'outside' fields before" (Verhoeven et al., 2016 - pag 714). The same logic can be applied to the originality index; having backward citations spread across many technological fields does not imply novelty in the search space because other patents might have relied on a similar knowledge recombination process before.

To more accurately identify novelty in the search space of a patent, Verhoeven et al. (2016) develop the "Novelty in Technology Origins" indicator. ¹² This indicator identifies a patent "as having Novelty in Technological Origins (NTO) if it makes a combination between its own IPC code and an IPC code from its referenced patents that has not occurred in the years previous to the application year of the patent" (Verhoeven et al., 2016 - pag 711) ¹³. The NTO indicator takes value one if the patent is classified as having novelty in technological origins and zero otherwise. An updated version of the data used in Verhoeven et al. (2016), computed using PATSTAT (2018), was made available by the authors. This being the case, to avoid possible truncation effects, the sample when using this variable is restricted to the period 1990-2014.

 $^{^{12}}$ The paper by Verhoeven et al. (2016) also introduces other indicators which are not discussed here because they are less relevant for this analysis.

¹³The IPC codes used to build the indicator are 6-digits IPC codes

The expected effect of electricity liberalisation on the NTO indicator is harder to describe than in the case of the radicalness and the originality indexes. If electricity liberalisation has an impact on the search space of clean-energy patents, it seems natural to expect that these patents will end up borrowing knowledge from, or expand into, the same technological fields. This is a direct consequence of the fact that only a limited number of technological fields will contain knowledge that is useful for the development of clean-energy technologies. That being said, the correlation between electricity liberalisation and the NTO indicator might take the form of an an inverted-U. When the market is first opened, we expect an increase in the number of novel connections made by clean-energy patents that are exploring outside fields not cited before. Further reforms of the electricity market might not trigger additional novel connections because the most interesting technological fields have already been explored. Section 5.2 presents evidence of this pattern and discusses how it relates with the main results of the paper and how it can help us interpreting them.

4 Methodology and Identification Strategy

To investigate the effect of electricity liberalisation on the indicators of interest I rely on the specification presented in equation (3). Note that, since the unit of analysis is the single patent application, the resulting dataset is a cross-section of patents; see Barbieri et al., (2020) and Rizzo et al., (2018) for similar applications.

$$PatInd_{i} = \beta_{1}LagPMR_{elec,i} + \beta_{2}Lag\mathbf{X}_{i} + \beta_{3}\mathbf{A}_{i} + App_{i} + Tech_{i} + Country_{i} + Year_{i} + \varepsilon_{i}$$
(3)

 $PatInd_i$ represents one of the patent indicators discussed in the previous section computed for patent *i*. The variable PMR_{elec} and the control variables in matrix **X** are lagged one year to account for the lag in the effect of policy variables.

The main coefficient of interest is β_1 , which quantifies the effect of a change in the degree of electricity liberalisation on $PatInd_i$.

The matrix **X** contains country-level control variables that might affect the development of clean-energy patents. First, I use the sub-index of the OECD Environmental Policy Stringency database (2022) that measures the use of policies aimed at supporting clean-energy innovations (Kruse et al., 2022). I also control for GDP per capita ¹⁴ and

 $^{^{14}\}mathrm{Source:}$ World Development Indicators - World Bank. Retrived through Our Wolrd in Data 22 June 2022

the price of crude oil imports computed at the country level ¹⁵. The inclusion of crude oil prices in the model is important in light of the effect that oil price shocks had on low-carbon patenting in the early 1980s (on this see for instance Calel and Dechezleprêtre., 2016).

A is a matrix of applicant and application level control variables. To proxy the resources and knowledge available at the applicant level I include in the regression the stock of clean-energy patent families computed for each applicant. The latter is built using the perpetual inventory method with a discount rate of 15% as discussed in the previous section. The relationship between the stock of clean-energy patent families and the indicator of interest is unlikely to be a linear one. On the one hand, new entrants are expected to develop more radical inventions and this would suggest a negative effect of the applicant's patent family stock on the radicalness and originality indexes. On the other hand, this negative effect might be weaker (or become positive) for applicants filing a lot of patents, because these actors will likely have access to more resources and can build on a larger body of knowledge. This being the case I also include in the specification the squared value of the applicant's patent family stock.

Following previous literature, application-level control variables are chosen based on how the patent indicators are built. First, since all the indicators we have discussed rely on information about prior knowledge, I control for the number of backward citations (Barieri et al., 2020, Hall et al., 2001)¹⁶. Furthermore, when the dependent variable is the radicalness index, I also control for the number of IPC full-digit codes the invention is allocated to, i.e. the scope of the patent (Barieri et al., (2020), Sapsalis et al., 2006). Results from previous literature suggests that the number of backward citations is positively correlated with our dependent variables, while the scope of the patent is negatively correlated with the radicalness index (see for instance, Barieri et al., 2020). Finally, I control for the number of applicants listed in the patent as more than one applicant working on the same invention could translate into more resources available for its development and a wider knowledge base.

The specification is then augmented with applicant (App_i) , technology class $(Tech_i)$, country $(Country_i)$ and year $(Year_i)$ fixed effects ¹⁷ The inclusion of applicant fixed

¹⁵Source: OECD (2022), Crude oil import prices (indicator). doi: 10.1787/9ee0e3ab-en (Accessed on 22 June 2022)

 $^{^{16}\}mathrm{To}$ deal with outliers in backward citations I follow Squicciarini et al (2013) and wisorize this variable over its 98% distribution

 $^{^{17}}$ In the OECD Patent Quality database information on the technological fields of the patent is based

effects in the model allows to control for time-invariant heterogeneity across applicants. However, at the same time it prevents the inclusion in the regression of the 6,059 patents from applicants that have filed only one patent family in the period of interest (Correia, 2015); see Table 2. This being the case, I will present the results both with and without applicants fixed effects in the model.

As the "treatment variable" (i.e., PMR_{elec}) is at the country level, I cluster the standard errors at this level (Abadie et al., 2017). Doing so generates few clusters (18) that are heterogeneous in size; see Table 1. In this context, wild cluster bootstrapping has been proved to perform much better than inference based on clustered standard errors relying on large-sample theory (Roodman et al 2019, Cameron and Miller, 2015). This being the case, for all specifications I report p-values and confidence intervals obtained implementing a wild cluster bootstrap using the STATA command 'boottest' (Roodman et al., 2019).

To estimate the model I rely on linear regression analysis ¹⁸. The radicalness and originality indexes can take any value between zero and one, thus a natural choice could have been to rely on a fractional model. Using a linear model allows me to easily include applicant fixed effects and, more importantly, to rely on wild-bootstrap based inference; which has been shown to be more reliable with respect to the alternative "score bootstrap" used for extremum estimators such as maximum likelihood (Roodman et al., 2019). Similar choices are not uncommon in the literature, see for instance Porter and Serra (2019). As robustness check, Section A.3 of the Appendix presents the results using a fractional probit model with inference based on score bootstrap. The same logic explains why I decided to use a Linear Probability Model when focusing on the NTO indicator as dependent variable of the model.

4.1 Instrumental Variable strategy

The proposed estimation strategy could suffer from endogeneity coming from different sources.

First, while the PMR_{elec} index is commonly used in the literature (e.g. Marino et al., 2019; Nicolli and Vona, 2019; Nicolli and Vona, 2016; Nesta et al., 2014), it is at

on the WIPO taxonomy (Schmoch, 2008). For patents allocated to more than one technology field they keep only the one with the majority of IPC codes. Finally, in case a patent has the same number of IPC codes for different technology fields it is randomly allocated to a technology fields.

¹⁸The model is estimated in STATA using the command reghdfe by Correia (2016)

best an imperfect proxy for the effective market power of incumbents (Nicolli and Vona, 2019).

Second, the development in the 1990s of more scalable technologies for energy generation (e.g., gas-fired plants and RETs) was a key factor that made the liberalisation of the electricity sector possible in the first place, so one might worry about possible reverse causality (Batlle and Ocaña, 2013).

Finally, countries that have the potential to develop more radical clean-energy innovations might also be the ones that reform first (or more) the electricity market. This could happen if "green lobbies" have the ability to affect both the development of radical clean-energy technologies and the regulation of the electricity sector (Nicolli and Vona 2019).

To deal with these issues, and uncover the causal effect of interest, I follow Nicolli and Vona (2019) and use an instrumental variable strategy where regulation in telecommunication is used as an instrument for regulation in electricity. Regulation in the telecommunication sector is measured using the PMR index for this particular industry. A robustness check using regulation in air transport as IV is provided in Section A.5 of the Appendix ¹⁹.

Both the reforms of telecommunication and air transport took place before the liberalisation of the electricity sector and played an instrumental role in giving momentum to the latter (Nicolli and Vona, 2019; Pollitt, 2012; Joskow, 2008). At the same time, these reforms can be considered independent from the lobbying power of actors in the energy sector and from technological developments in energy-supply technologies. On this assumptions rest the validity of the proposed strategy.

Wild bootstrap inference requires some caution when it comes to IV estimation. First, following Roodman et al. (2019) and Davidson and MacKinnon (2010) I rely on equal-tail p-values as opposed to symmetric p-values in order to assess the statistical significance of these estimates.²⁰. Second, the tests for weak instruments relying on first-stage regressions also requires additional attention when clustered standard errors relying on large-sample theory are applied in sub-optimal contexts. Young (2022) uses a sample of 1309 instrumental variables regressions in 30 published papers and shows that

¹⁹In the case of Air Transport, when using clustered standard errors based on asymptotic theory the instrument appears to be strong enough, but the p-value of the first-stage regression obtained using wild bootstrap would lead us to conclude the opposite. This being the case, it is possible that these suffer from weak-instruments issues

²⁰Results are however very similar using symmetric p-value

these tests reject the null 100% of the times at the 0.01 level using clustered standard errors based on large-sample theory, but only 80% of the times once bootstrap techniques are applied. This being the case, when testing the strength of the instrument I will also report the p-value from the first-stage regression obtained using wild bootstrap techniques. Third, the choice between studentized wild bootstrapping (bootstrapt-t) and unstudentized wild bootstrapping (bootstrap-c) is not straightforward in IV settings. Studentized wild bootstrapping is generally considered the best choice according to asymptotic theory (Hall, 1992) but results from Young (2022) and Wang (2021) provide evidence that in IV applications unstudentized wild bootstrapping (bootstrap-c) might perform better. In light of this, I will provide inference on the main coefficient of interest using also unstudentized wild bootstrapping.

5 Results

5.1 The effect of electricity liberalisation on the search space of clean-energy technologies

Table 5 shows the results from the naïve OLS estimation of the model discussed in Section 4. Note that the wild cluster bootstrap does not assume normality and therefore it does not calculate standard errors (Roodman et al 2019). This being the case I follow Porter and Serra (2019) and report the obtained p-values and 95% confidence intervals in the result tables.

The coefficient associated with PMR_{elec} is always statistically significant and of the expected sign. When the dependent variable is the radicalness index (columns 1 and 2) the estimated effect is significant at the 99% threshold, while for the originality index (columns 3 and 4) the threshold of significance is 95%. The magnitude of the estimated coefficient is much stronger in the case of the radicalness index than for the originality index. To see this, we can compare the estimated effect of a one-unit change in PMR_{elec} with the average value that these indexes take in the sample. The average value of the radicalness index among patent families in the sample is 0.32 (see Table 4). The estimated effect in columns 1 and 2 is therefore roughly 6% of the mean of the variable. The average value that the originality index takes in the sample is 0.69 (see Table 4). Thus, in this case, the effect of PMR_{elec} ranges between 0.6% and 1.1% of the mean of the variable. In other words, the estimated effect of electricity liberalisation on the originality index, even if statistically different from zero, is close to zero in magnitude.

Looking at the other control variables, we see that they are generally of the expected sign. ²¹ Country level control variables other than the PMR_{elec} are generally not significant, on the other hand applicants and application level controls display a strong correlation with both dependent variables. As expected, the coefficient associated with the scope of the patent is negative and highly significant and the one associated with backward citations is always positive and significant at least at the 95% threshold. Contrary to expectations, the number of applicants listed in the document is not an important factor in the analysis. Finally, there is evidence in favor of the hypothesised inverted-U relationship between the patent family stock and the radicalness and originality indexes.

Table 6 reports the result when implementing the IV strategy discussed in Section 4.1. The test for weak instrument based on the first-stage regression with clustered standard errors that rely on large-sample theory always returns an F statistic well above the usual cut-off level of 10 (Stock et al., 2002). Following Young (2022), I further test the strength of the instrument looking at the p-value obtained from the first-stage regression when testing the coefficient of the instrument using wild clustered bootstrapping. This p-value is always significantly lower that 0.01. As shown in Appendix A.4, this is not true for the robustness check done using air transport as IV, which however appears to be strong enough when the test is based on clustered standard errors that rely on large-sample theory.

Results from column 1 to 3 of Table 6 are in line with what we see in Table 5. In column 4 the p-value associated with the estimated coefficient for PMR_{elec} is substantially higher and the effect is no longer statistically significant. Looking at the magnitude of the estimated coefficients, 2SLS estimates always fall into the OLS 95% confidence interval, thus providing evidence in favor of OLS estimates. In light of this, the loss of significance in column 4 could be explained by the combination of the lower statistical power associated with 2SLS estimates and an estimated effect of PMR_{elec} on the originality index that is close to zero in magnitude.

Following Young (2022) and Wang (2021), I further test the significance of the PMR_{elec} coefficient relying on unstudentized wild bootstrapping (bootstrap-c). The p-value obtained applying this procedure confirms the significant effect of electricity liberalisation on the radicalness index and casts additional doubts on the relationship between electricity liberalisation and the originality index, with the coefficient of PMR_{elec}

 $^{^{21}}$ The only two exceptions are the estimated coefficients for the price of oil imports in column 4 and for the *Number of applicants* in column 1. Note however that in both cases these coefficients are far from being significant

in column 3 that is no longer significant at the 95% level.

Summing up, the estimated effect of electricity liberalisation on the radicalness index is always statistically significant and meaningful in magnitude. On the other hand, the effect of electricity liberalisation on the originality index is weak at best and not robust. This pattern is confirmed in the robustness checks presented in the Appendix (see Sections A.3, A.4, A.5 and A.6). When considered together, these results suggest that electricity liberalisation pushes clean-energy patents to cite knowledge from technological fields other than their own, but the bulk of clean-energy patents' search space remains concentrated around the same number of technological fields.

Overall, the results presented in this section are evidence that electricity liberalisation affects the search space of clean-energy patents and can be a driver for the development of radical clean-energy technologies, as sometimes hypothesised in the literature (Negro et al., 2012, Markard and Truffer, 2006). At the same time, the analysis also helps to define the boundaries of this effect by describing what features of clean-energy patents' search space are affected by higher levels of competition in the electricity market.

The next section will exploit the NTO indicator in order to further investigate the obtained results.

5.2 Electricity liberalisation and Novelty in Technological Origins: further insights on the main results

Table 7 and 8 provide evidence for the inverted-U correlation between the NTO indicator and electricity liberalisation hypothesised in Section 3.1.3. The results in column 1 of Table 7 are obtained using a linear probability model and inference based on clustered wild bootstrapping. The estimated model is the same presented in Section 4, without the inclusion of applicant fixed effects and with the addition of the squared value of the PMR_{elec} index. Columns 2 and 3 estimate the same model using as dependent variables the radicalness and originality indexes. In Table 8, I present the marginal effects of these estimates. The remainder of this section will discuss the three main takeaways that can be drawn from this analysis, focusing on how these results can inform our interpretation of the results presented in the previous section.

First, the expected inverted-U correlation between the PMR_{elec} index and the NTO indicator is borne out by the data. The liberalisation of an heavily regulated electricity market is correlated with an higher likelihood that clean-energy technologies will make novel connections in their search space. As the electricity market becomes more com-

petitive, this correlation first becomes not significant then changes sign. As discussed in Section 3.1.3, this pattern is expected because there are only a limited number of technological fields that have significant synergies with clean-energy technologies. When these fields have been explored, further liberalising the electricity market is unlikely to trigger a significant number of novel connections.

Second, column 2 displays the opposite trend, with higher level of PMR_{elec} that are correlated with a stronger effect of liberalisation on the radicalness index. Looking together at the results from column 1 and 2 of Table 8 we see that, as the electricity market becomes more competitive, the propensity of clean-energy technologies to cite new technological fields and their propensity to cite outside technological fields start moving in opposite directions. This suggests that, at first, the exploration of new technological fields can be a driver of the increase in outside knowledge cited by clean-energy patents. However, after a certain point, this is no longer the case; electricity liberalisation still drives clean-energy technologies to cite outside technological fields, but these are the same fields already explored when the market was first opened to competition.

Finally, column 3 confirms that the effect on the originality index is weak and characterised by a low level of significance. However, if we look only at the magnitude of the estimated marginal effect, it is interesting to note that it decreases monotonically as the market becomes more liberalised, following the same trend as the NTO indicator.

	(1)	(2)	(3)	(4)
	Rad	Rad	Ori	Ori
Lag PMR_{elec}	-0.0190	-0.0197	-0.0042	-0.0079
	(0.0002)	(0.0000)	(0.0303)	(0.0502)
	[-0.0237, -0.0157]	[-0.0268, -0.0145]	[-0.0098, -0.0010]	[-0.0126, 0.0000]
Lag EPS_{tech}	0.0022	0.0015	0.0042	0.0025
	(0.4151)	(0.6101)	(0.1650)	(0.3446)
	[-0.0054, 0.0069]	[-0.0072, 0.0064]	[-0.0025, 0.0100]	[-0.0049, 0.0079]
Lag GDP_{pc}	0.0628	0.1353	0.324	0.3769
	(0.6006)	(0.1219)	(0.1183)	(0.0306)
	[-0.3360, 0.3514]	[-0.0506, 0.3325]	[-0.0613, 0.5497]	[0.0405, 0.7544]
Lag Oil price (imports)	0.1442	0.0993	0.0054	-0.0227
	(0.1981)	(0.5052)	(0.9587)	(0.8238)
	[-0.0770, 0.3785]	[-0.1266, 0.4299]	[-0.2808, 0.2190]	[-0.3772, 0.2396]
Patent Scope	-0.0263	-0.0192		
	(0.0014)	(0.0002)		
	[-0.0445, -0.0140]	[-0.0311, -0.0080]		
Bwd Citations	0.0031	0.0038	0.0101	0.0111
	(0.0008)	(0.0011)	(0.0340)	(0.0025)
	[0.0014, 0.0062]	[0.0020, 0.0065]	[0.0068, 0.0156]	[0.0087, 0.0147]
Number Applicants	-0.0008	0.0018	0.0066	0.0080
	(0.8266)	(0.6482)	(0.1764)	(0.1963)
	[-0.0258, 0.0124]	[-0.0129, 0.0129]	[-0.0085, 0.0167]	[-0.0050, 0.0115]
Family Stock	-0.0142	-0.0286	-0.0128	-0.0333
	(0.0605)	(0.0150)	(0.0028)	(0.0808)
	[-0.0330, 0.0473]	[-0.0357, -0.0093]	[-0.0321, -0.0044]	[-0.0504, 0.0028]
Family Stock ²	0.0021	0.0041	0.0028	0.0049
	(0.0547)	(0.0095)	(0.0025)	(0.0866)
	[-0.0002, 0.0039]	[0.0016, 0.0057]	[0.0006, 0.0061]	[-0.0011, 0.0084]
Applicant FE	Yes	No	Yes	No
Technology FE	Yes	Yes	Yes	Yes
Country FE	Yes	Yes	Yes	Yes
Year FE	Yes	Yes	Yes	Yes
Observations	49,583	55,642	49,542	55,600

Table 5: OLS estimates of Equation (3)

Notes: OLS regressions. I report wild bootstrap cluster p-values in parentheses and wild bootstrap cluster 95% confidence intervals in square brackets, generated using boottest command in Stata (Roodman et al., 2019) for standard errors clustered at the country level (18 clusters) 23

	(1)	(2)	(3)	(4)
	Rad	Rad	Ori	Ori
Lag PMR_{elec}	-0.0221	-0.0169	-0.0043	-0.0054
	(0.0012)	(0.0050)	(0.0370)	(0.2498)
	[-0.0359, -0.0146]	[-0.0237, -0.0090]	[-0.0211, -0.0002]	[-0.01529, 0.0089]
Lag EPS_{tech}	0.0027	0.0012	0.0042	0.0022
	(0.3762)	(0.6883)	(0.1186)	(0.4206)
	[-0.0058, 0.0083]	[-0.0067, 0.0063]	[-0.0017, 0.0090]	[-0.0062, 0.0079]
Lag GDP_{pc}	0.0303	0.1572	0.3300	0.3970
	(0.7789)	(0.1606)	(0.1904)	(0.0372)
	[-0.3787, 0.3365]	[-0.0945, 0.4251]	[-0.1517, 0.5489]	[0.0323, 0.8186]
Lag Oil price (imports)	0.1563	0.0856	0.0056	-0.0353
	(0.1656)	(0.6083)	(0.9729)	(0.7061)
	[-0.0646, 0.3974]	[-0.1374, 0.4344]	[-0.2808, 0.2243]	[-0.3859, 0.2111]
Patent Scope	-0.0262	-0.0192		
	(0.0010)	(0.0000)		
	[-0.0443, -0.0140]	[-0.0311, -0.0080]		
Bwd Citations	0.0031	0.0038	0.0101	0.0111
	(0.0006)	(0.0010)	(0.0324)	(0.0016)
	[0.0014, 0.0062]	[0.0020, 0.0064]	[0.0068, 0.016]	[0.0086, 0.0147]
Number Applicants	-0.0008	0.0019	0.0066	0.0080
	(0.8289)	(0.6457)	(0.1776)	(0.1924)
	[-0.0260, 0.0123]	[-0.01275, 0.01290]	[-0.0090, 0.0168]	[-0.0047, 0.0115]
Family Stock	-0.0142	-0.0287	-0.0128	-0.0333
	(0.0656)	(0.0136)	(0.0100)	(0.0786)
	[-0.0324, 0.0477]	[-0.0358, -0.0092]	[-0.0322, -0.0026]	[-0.0503, 0.0025]
Family Stock ²	0.0021	0.0041	0.0028	0.0050
	(0.0576)	(0.0094)	(0.0040)	(0.0866)
	[-0.0003, 0.0040]	[0.0016, 0.0057]	[0.0005, 0.0060]	[-0.0009, 0.0088]
Applicant FE	Yes	No	Yes	No
Other FEs	Yes	Yes	Yes	Yes
P-value Bootstrap-c PMR	0.0000	0.0010	0.0620	0.2400
F-stat first stage	57.05	75.22	57.08	75.27
First Stage Bootstrap	0.0026	0.0015	0.0026	0.0015
Observations	49,583	55,642	49,542	55,600

Table 6: 2SLS estimates of Equation (3)

Notes: 2SLS regressions. Regulation in the telecommunication sector is used as instrument for regulation in the electricity sector. I report wild bootstrap cluster p-values in parentheses and wild bootstrap cluster 95% confidence intervals in square brackets, generated using boottest command in Stata (Roodman et al., 2019) for standard errors clustered at the country level (18 clusters). The significance of the coefficient associated with PMR_{elec} is tested also relying on unstudentized wild bootstrapping (P-value Bootstrap-c PMR_{elec})

	(1)	(2)	(3)
	NTO	Rad	Ori
Lag PMR_{elec}	0.03618	-0.0258	-0.0026
	(0.0082)	(0.0040)	(0.7677)
	[0.0145, 0.0092]	[-0.0421, -0.0145]	[-0.0258, 0.0212]
${\rm Lag}\;{\rm PMR^2}_{elec}$	-0.0072	0.0011	-0.0009
	(0.0002)	(0.0994)	(0.4638)
	[-0.0165, -0.0035]	[-0.0005, 0.0033]	[-0.0044, 0.0027]
Controls	Yes	Yes	Yes
Applicant FE	No	No	No
Technology FE	Yes	Yes	Yes
Country FE	Yes	Yes	Yes
Year FE	Yes	Yes	Yes
Observations	45,574	55,642	55,600

Table 7: OLS estimates of Equation (3) with the addition of Lag PMR^2_{elec}

Notes: OLS regressions of model 3 with the addition of Lag PMR_{elec}^2 . I report wild bootstrap cluster p-values in parentheses and wild bootstrap cluster 95% confidence intervals in square brackets, generated using boottest command in Stata (Roodman et al., 2019) for standard errors clustered at the country level (18 clusters).

	(1)	(2)	(3)
	NTO	Rad	Ori
Lag $PMR_{elec} = 0$	0.0362	-0.0258	-0.0026
	(0.0082)	(0.0040)	(0.7677)
	[0.0145, 0.0917]	[-0.0421, -0.0145]	[-0.0258, 0.0212]
Lag $PMR_{elec} = 1$	0.0219	-0.0236	-0.0045
	(0.0072)	(0.0016)	(0.4825)
	[0.0069, 0.065]	[-0.0360, -0.0147]	[-0.0203, 0.0119]
Lag $PMR_{elec} = 2$	0.0076	-0.0214	-0.0064
	(0.1082)	(0.0004)	(0.1822)
	[-0.0019, 0.0285]	[-0.0302, -0.0142]	[-0.0152, 0.0051]
Lag $PMR_{elec} = 3$	-0.0067	-0.0192	-0.0083
	(0.2892)	(0.0000)	(0.0181)
	[-0.0167, 0.0130]	[-0.0262, -0.0139]	[-0.0120, -0.0030]
Lag $PMR_{elec} = 4$	-0.0210	-0.0170	-0.0102
	(0.0352)	(0.0004)	(0.0323)
	[-0.0523, -0.0018]	[-0.0228, -0.0123]	[-0.0147, -0.0030]
Lag $PMR_{elec} = 5$	-0.0353	-0.0148	-0.0122
	(0.0044)	(0.0026)	(0.0703)
	[-0.0880, -0.0150]	[-0.0215, -0.0090]	[-0.0205, 0.0037]
Lag $PMR_{elec} = 6$	-0.0496	-0.0126	-0.0141
	(0.0019)	(0.0196)	(0.1521)
	[-0.1193, -0.0238]	[-0.0216, -0.0038]	[-0.0283, 0.0096]
Controls	Yes	Yes	Yes
Applicant FE	No	No	No
Other FEs	Yes	Yes	Yes
Observations	45,574	55,642	55,600

Table 8: Marginal Effects of models estimated in Table 7

Notes: The table reports marginal effect of PMR_{elec} for the models presented in Table 7. I report wild bootstrap cluster p-values in parentheses and wild bootstrap cluster 95% confidence intervals in square brackets, generated using boottest command in Stata (Roodman et al., 2019) for standard errors clustered at the country level (18 clusters)

6 Discussion and conclusions

The literature on the relationship between electricity liberalisation and innovation has sometimes hypothesised that a more competitive electricity market can affect the search space of clean-energy patents and therefore be a driver for the development of radical clean-energy technologies (Negro et al., 2012, Makard and Truffer, 2006). This paper contributes to this literature by being the first one to test this idea empirically using patent data and patent-level indicators (see, e.g., Verhoeven et al,. 2016, Squicciarini et al 2013).

The results suggest that electricity liberalisation pushes clean-energy patents to explore "outside" technological fields, i.e. to borrow knowledge from technological fields other than their own. This pattern is correlated with the radicalness of an invention, as it signals that the patent relies on different paradigms with respect to the one to which it is applied (Barbieri et al 2020, Squicciarini et al 2013, Shane 2001). At the same time, electricity liberalisation does not significantly change the breath of clean-energy patents' search space; a characteristic that is correlated with the originality and complexity of the invention (Barbieri et al., 2020, Dechezleprêtre et al., 2017; Verhoeven et al., 2016, Squicciarini et al. 2013, Trajtenberg et al., 1997).

More insights on the relationship of interest can be drawn by looking at the correlation between electricity liberalisation and an indicator of novelty in the search space of patents (Verhoeven et al., 2016). The liberalisation of an heavily regulated electricity market is positively correlated with the likelihood that clean-energy patents will explore new technological fields, i.e. technological fields never cited before by patents in the same domain. As the electricity market becomes more competitive, this is no longer the case. This patter is expected, due to the finite number of technological fields from which clean-energy technologies can borrow useful knowledge. Once these fields have been explored, further liberalising the electricity market is unlikely to trigger a substantial numbers of new explorations. On the contrary, the positive effect of electricity liberalisation on clean-energy patents' propensity to cite knowledge from outside (but not necessarily new) technological fields remains strong and statistically significant as the market becomes more competitive. This suggest that, when the market is first opened to competition, the exploration of new technological fields can be a driver of clean-energy patents' increased reliance on outside knowledge. On the other hand, additional reforms of an already liberalised electricity market increase the propensity of clean-energy technologies to cite outside technological fields, but these fields are the same already

explored when the market was first opened to competition.

The results are therefore consistent with the claim that electricity liberalisation is a driver for radical clean-energy technologies. At the same time, by describing precisely how a more competitive electricity market affects the search space of clean-energy technologies, they also point out the limits of this effect.

The main contribution of the paper is to describe in detail the effect of electricity liberalisation on the search space of clean-energy technologies. This allows a more precise definition of how, and to what extent, electricity liberalisation can be a driver for the development of radical clean-energy technologies. From a policy making standpoint, understanding the relationship between electricity regulation and the quality of cleanenergy innovations is particularly important in the current situation. To deal with the recent power price crisis, various changes to the design of electricity markets in Europe are under discussion and some of the proposed ideas would likely affect the level of competition in the market if implemented (ACER 2022). In this regard, these results warns about the possible negative effects of changes in the electricity market design that weaken competition on the quality of clean energy technologies.

The analysis has some limitations, which could be addressed by further research. First, the United States, which are a major developer of clean-energy technologies, could not be included in the sample. In part, this is because the 2020 iteration of the PMR_{elec} index does not include the U.S. at the time of writing (Vitale et al., OECD, 2020). More importantly however, a similar analysis for the U.S. would need to be carried out at the state level rather than the federal one and thus would require an indicator of regulation in electricity computed for each state. The use of an indicator computed at the federal level would hide the heterogeneous regulatory environments to which inventors in the different states are exposed. This heterogeneity is significant because the U.S. has never enacted a mandatory federal restructuring law, leaving to the states the most important decisions (Joskow, 2008). A similar analysis for the U.S., carried out at the state level, is therefore left for future research.

Second, the analysis takes only an ex-ante approach to the definition of radicalness (Verhoeven et al. 2016). This is the natural consequence of the research question of the paper, which is concerned with the effect of electricity liberalisation on the search space of clean-energy technologies. In future research, the analysis could be expanded by taking also an ex-post approach to the definition of radical technologies (see for instance Barbieri et al 2020, Acemoglue et al 2020).

References

Abadie, A., Athey, S., Imbens, G. W., Wooldridge, J. (2017). When should you adjust standard errors for clustering? (No. w24003). National Bureau of Economic Research. DOI: https://doi.org/10.3386/w24003

Acemoglu, D., Akcigit, U., Celik, M. A. (2022). Radical and incremental innovation: The roles of firms, managers, and innovators. American Economic Journal: Macroeconomics, 14(3), 199-249. DOI: https://doi.org/10.1257/mac.20170410

Agency for the Cooperation of Energy Regulators (ACER) - "ACER's Final Assessment of the EU Wholesale Electricity Market Design" - (2022). Link: https://rb.gy/jzbif6

Akcigit, U., Kerr, W. R. (2018). Growth through heterogeneous innovations. Journal of Political Economy, 126(4), 1374-1443. DOI: https://doi.org/10.1086/697901

Arthur, W. B. (2007). The structure of invention. Research policy, 36(2), 274-287. DOI: https://doi.org/10.1016/j.respol.2006.11.005

Barbieri, N., Marzucchi, A., Rizzo, U. (2020). Knowledge sources and impacts on subsequent inventions: Do green technologies differ from non-green ones?. Research Policy, 49(2), 103901. DOI: https://doi.org/10.1016/j.respol.2019.103901

Batlle, C., Ocaña, C. (2013). Electricity regulation: principles and institutions. In Regulation of the power sector (pp. 125-150). Springer, London.

Batlle, C., Schittekatte, T., Knittel, C. R. (2022). Power price crisis in the EU 2.0+: Desperate times call for desperate measures. Available at SSRN 4074014. DOI: http://dx.doi.org/10.2139/ssrn.4074014

Baudry, M., Dumont, B. (2006). Comparing firms' triadic patent applications across countries: Is there a gap in terms of RD effort or a gap in terms of performances?. Research Policy, 35(2), 324-342. DOI: https://doi.org/10.1016/j.respol.2005.12.004

Calel, R., Dechezleprêtre, A. (2016). Environmental policy and directed technological change: evidence from the European carbon market. Review of economics and statistics, 98(1), 173-191. DOI: https://doi.org/10.1162/REST_a_00470

Carley, S. (2009). State renewable energy electricity policies: An empirical evaluation of effectiveness. Energy policy, 37(8), 3071-3081. DOI: https://doi.org/10.1016/j.enpol.2009.03.062

Conner, K. R. (1988). Strategies for product cannibalism. Strategic Management Journal, 9(S1), 9-26. DOI: https://doi.org/10.1002/smj.4250090704

Correia, S. (2015). Singletons, cluster-robust standard errors and fixed effects: A bad mix. Technical Note, Duke University, 7. Link: http://scorreia.com/research/singletons.pdf

Correia, S. (2016, August). reghdfe: Estimating linear models with multi-way fixed effects. In 2016 Stata Conference (No. 24). Stata Users Group. Link: http://fmwww.bc.edu/repec/chic2016/chicago16_correia.pdf

Davidson, R., MacKinnon, J. G. (2010). Wild bootstrap tests for IV regression. Journal of Business Economic Statistics, 28(1), 128-144. DOI: https://doi.org/10.1198/jbes.2009.07221

De Marchi, V. (2012). Environmental innovation and RD cooperation: Empirical evidence from Spanish manufacturing firms. Research policy, 41(3), 614-623. DOI: https://doi.org/10.1016/j.respol.2011.10.002

Delmas, M. A., Montes-Sancho, M. J. (2011). US state policies for renewable energy: Context and effectiveness. Energy Policy, 39(5), 2273-2288. DOI: https://doi.org/10.1016/j.enpol.2011.01.034

Dooley, J. J. (1998). Unintended consequences: energy RD in a deregulated energy market. Energy Policy, 26(7), 547-555. DOI: https://doi.org/10.1016/S0301-4215(97) 00166-3

Dolphin, G., Pollitt, M. (2020). Identifying Innovative Actors in the Electricicity Supply Industry Using Machine Learning: An Application to UK Patent Data. Link: https://www.jstor.org/stable/pdf/resrep30421.pdf

Kafouros, M., Aliyev, M., Krammer, S. M. (2021). Do firms profit from patent litigation? The contingent roles of diversification and intangible assets. Research Policy, 50(6), 104263. DOI: https://doi.org/10.1016/j.respol.2021.104263

Kruse, T., Dechezleprêtre, A., Saffar, R., Robert, L. (2022). Measuring environmental policy stringency in OECD countries: An update of the OECD composite EPS indicator. DOI: https://doi.org/10.1787/90ab82e8-en

Hall, B. H. (2005). Measuring the Returns to RD: The Depreciation Problem. Annales d'Économie et de Statistique, 79/80, 341–381. DOI: https://doi.org/10.2307/ 20777581

Hall, B. H., Jaffe, A. B., Trajtenberg, M. (2001). The NBER patent citation data file: Lessons, insights and methodological tools. DOI: https://doi.org/10.3386/w8498

Hall, P. (2013). The bootstrap and Edgeworth expansion. Springer Science Business Media. Link: https://people.eecs.berkeley.edu/~jordan/sail/readings/edgeworth.pdf

Hughes, T.P. (1987). The evolution of large technological systems. In: Bijker, W., Hughes, T.P., Pinch, T. (Eds.), The Social Construction of Technological Systems. Cambridge, MA, pp. 51-82. Link https://rb.gy/gecjk6

Hussinger, K., Pacher, S. (2019). Information ambiguity, patents and the market value of innovative assets. Research policy, 48(3), 665-675. DOI: https://doi.org/10.1016/j.respol.2018.10.022

Young, A. (2022). Consistency without inference: Instrumental variables in practical application. European Economic Review, 104112. DOI: https://doi.org/10.1016/j.euroecorev.2022.104112

Jacobsson, S., Bergek, A. (2004). Transforming the energy sector: the evolution of technological systems in renewable energy technology. Industrial and corporate change, 13(5), 815-849. DOI: https://doi.org/10.1093/icc/dth032

Jamasb, T., Pollitt, M. (2008). Liberalisation and RD in network industries: The case of the electricity industry. Research Policy, 37(6-7), 995-1008. DOI: https://doi.org/10.1016/j.respol.2008.04.010

Jamasb, T., Pollitt, M. G. (2011). Electricity sector liberalisation and innovation: An analysis of the UK's patenting activities. Research Policy, 40(2), 309-324. DOI: https://doi.org/10.1016/j.respol.2010.10.010

Joerges, B. (1996). Large technical systems and the discourse of complexity. Complex technical systems, 55-72. Link: https://rb.gy/h6pop1

Joskow, P. L. (2008). Lessons learned from electricity market liberalization. The Energy Journal, 29(Special Issue 2). DOI: DOI:10.5547/ISSN0195-6574-EJ-Vol29-NoSI2-3

Klepper, S. (1996). Entry, exit, growth, and innovation over the product life cycle. The American economic review, 562-583. Link https://www.jstor.org/stable/2118212

Li, D., Heimeriks, G., Alkemade, F. (2020). The emergence of renewable energy technologies at country level: relatedness, international knowledge spillovers and domestic energy markets. Industry and Innovation, 27(9), 991-1013. DOI: https://doi.org/10.1080/13662716.2020.1713734

Marino, M., Parrotta, P., Valletta, G. (2019). Electricity (de) regulation and innovation. Research policy, 48(3), 748-758. DOI: https://doi.org/10.1016/j.respol. 2018.11.005

Markard, J., Truffer, B. (2006). Innovation processes in large technical systems: Market liberalization as a driver for radical change?. Research policy, 35(5), 609-625. DOI: https://doi.org/10.1016/j.respol.2006.02.008

Negro, S. O., Alkemade, F., Hekkert, M. P. (2012). Why does renewable energy diffuse

so slowly? A review of innovation system problems. Renewable and sustainable energy reviews, 16(6), 3836-3846. DOI: https://doi.org/10.1016/j.rser.2012.03.043

Nelson, R. R. (1985). An evolutionary theory of economic change. harvard university press. Link: https://www.hup.harvard.edu/catalog.php?isbn=9780674272286

Nesta, L., Vona, F., Nicolli, F. (2014). Environmental policies, competition and innovation in renewable energy. Journal of Environmental Economics and Management, 67(3), 396-411. DOI: https://doi.org/10.1016/j.jeem.2014.01.001

Nicolini, M., Tavoni, M. (2017). Are renewable energy subsidies effective? Evidence from Europe. Renewable and Sustainable Energy Reviews, 74, 412-423. DOI: https://doi.org/10.1016/j.rser.2016.12.032

Nicolli, F., Vona, F. (2016). Heterogeneous policies, heterogeneous technologies: The case of renewable energy. Energy Economics, 56, 190-204. DOI: https://doi.org/10.1016/j.eneco.2016.03.007

Nicolli, F., Vona, F. (2019). Energy market liberalization and renewable energy policies in OECD countries. Energy Policy, 128, 853-867. DOI: https://doi.org/10.1016/j. enpol.2019.01.018

Olsson, O. (2000). Knowledge as a set in idea space: An epistemological view on growth. Journal of economic growth, 5(3), 253-275. DOI: https://doi.org/10.1023/A:1009829601155

Pollitt, M. G. (2012). The role of policy in energy transitions: Lessons from the energy liberalisation era. Energy policy, 50, 128-137. DOI: https://doi.org/10.1016/j.enpol.2012.03.004

Porter, C., Serra, D. (2020). Gender differences in the choice of major: The importance of female role models. American Economic Journal: Applied Economics, 12(3), 226-54. DOI: https://doi.org/10.1257/app.20180426

Reinganum, J. F. (1983). Uncertain innovation and the persistence of monopoly. The

American Economic Review, 73(4), 741-748. Link: https://www.jstor.org/stable/ 1816571

Rizzo, U., Barbieri, N., Ramaciotti, L., Iannantuono, D. (2020). The division of labour between academia and industry for the generation of radical inventions. The Journal of Technology Transfer, 45(2), 393-413. DOI: https://doi.org/10.1007/s10961-018-9688-y

Roodman, D., Nielsen, M. Ø., MacKinnon, J. G., Webb, M. D. (2019). Fast and wild: Bootstrap inference in Stata using boottest. The Stata Journal, 19(1), 4-60. DOI: https://doi.org/10.1177/1536867X1983087

Rosenkopf, L., Nerkar, A. (2001). Beyond local search: boundary-spanning, exploration, and impact in the optical disk industry. Strategic management journal, 22(4), 287-306. DOI: https://doi.org/10.1002/smj.160

Sanyal, P., Ghosh, S. (2013). Product market competition and upstream innovation: evidence from the US electricity market deregulation. Review of Economics and Statistics, 95(1), 237-254. DOI: https://doi.org/10.1162/REST_a_00255

Sterlacchini, A. (2012). Energy RD in private and state-owned utilities: An analysis of the major world electric companies. Energy policy, 41, 494-506. DOI: https://doi.org/10.1016/j.enpol.2011.11.010

Shane, S. (2001). Technological opportunities and new firm creation. Management science, 47(2), 205-220. DOI: https://doi.org/10.1287/mnsc.47.2.205.9837

Sapsalis, E., de la Potterie, B. V. P., Navon, R. (2006). Academic versus industry patenting: An in-depth analysis of what determines patent value. Research Policy, 35(10), 1631-1645. DOI: https://doi.org/10.1016/j.respol.2006.09.014

Schmoch, U. (2008). Concept of a technology classification for country comparisons. Final report to the world intellectual property organisation (wipo), WIPO. Link: http:// www.world-intellectual-property-organization.com/edocs/mdocs/classifications/ en/ipc_ce_41/ipc_ce_41_5-annex1.pdf Squicciarini, M., Dernis, H., Criscuolo, C. (2013). Measuring patent quality: Indicators of technological and economic value. DOI: https://doi.org/10.1787/5k4522wkw1r8-en

Trajtenberg, M., Henderson, R., Jaffe, A. (1997). University versus corporate patents: A window on the basicness of invention. Economics of Innovation and new technology, 5(1), 19-50. DOI: https://doi.org/10.1080/10438599700000006

Vitale, C., Bitetti, R., Wanner, I., Danitz, E., Moiso, C. (2020). The 2018 edition of the OECD PMR indicators and database: Methodological improvements and policy insights. DOI: https://doi.org/10.1787/2cfb622f-en

Wang, W. (2021). Wild bootstrap for instrumental variables regression with weak instruments and few clusters. Link: https://mpra.ub.uni-muenchen.de/106227/

Wang, N., Mogi, G. (2017). Deregulation, market competition, and innovation of utilities: Evidence from Japanese electric sector. Energy Policy, 111, 403-413. DOI: https://doi.org/10.1016/j.enpol.2017.09.044

Winter, S. G. (1984). Schumpeterian competition in alternative technological regimes. Journal of Economic Behavior Organization, 5(3-4), 287-320. DOI: https://doi.org/10.1016/0167-2681(84)90004-0

A Appendix

A.1 Y02E technologies

Y02E / 10: Energy generation through renewable energy sources:

- Geothermal energy
- Hydro energy
- Energy from the sea, e.g. using wave energy or salinity gradient
- Solar thermal energy, e.g. solar towers
- Photovoltaic [PV] energy
- Thermal-PV hybrids
- Wind energy

Y02E / 20: Combustion technologies with mitigation potential:

- Heat utilisation in combustion or incineration of waste
- Combined heat and power generation [CHP]
- Combined cycle power plant [CCPP], or combined cycle gas turbine [CCGT]
- Technologies for a more efficient combustion or heat usage
- Direct CO2 mitigation
- Indirect CO2mitigation, i.e. by acting on non CO2directly related matters of the pro-

cess, e.g. pre-heating or heat recovery

Y02E / 30: Energy generation of nuclear origin:

- Nuclear fusion reactors
- Nuclear fission reactors

Y02E / 40: Technologies for an efficient electrical power generation, transmission or distribution:

- Flexible AC transmission systems [FACTS]
- Active power filtering [APF]
- Reactive power compensation
- Arrangements for reducing harmonics
- Arrangements for eliminating or reducing asymmetry in polyphase networks
- Superconducting electric elements or equipment; Power systems integrating superconducting elements or equipment
- Smart grids as climate change mitigation technology in the energy generation sector

Y02E / 50: Technologies for the production of fuel of non-fossil origin:

• Biofuels, e.g. bio-diesel

• Fuel from waste, e.g. synthetic alcohol or diesel

Y02E / 60: Enabling technologies; Technologies with a potential or indirect contribution to GHG emissions mitigation

- Energy storage using batteries
- Energy storage using capacitors
- Thermal energy storage
- Mechanical energy storage, e.g. flywheels or pressurised fluids
- Hydrogen technology
- Smart grids in the energy sector

Y02E / 70 : Other energy conversion or management systems reducing GHG emission



A.2 PMR_{elec} index by Country and year





A.3 Fractional Probit Model

	(1)	(2)
	Radicalness	Originality
Lag PMR_{elec}	-0.0187	-0.0086
	(0.0002)	(0.0372)
Controls	Yes	Yes
Applicant FE	No	No
Technology FE	Yes	Yes
Country FE	Yes	Yes
Year FE	Yes	Yes
Yes		
Observations	$55,\!642$	$55,\!600$

Notes: Fractional probit regressions. The table reports marginal effects as opposed to regression coefficients. Score bootstrap cluster p-values in parentheses are generated using boottest command in Stata (Roodman et al., 2019) for standard errors clustered at the country level (18 clusters). The estimated model is the model presented in Section 4, without the inclusion of applicant fixed effects., i.e. the same model estimated in column 2 and 4 of Table 5.

	(1)	(2)	(3)	(4)
	Rad	Rad	Ori	Ori
Lag PMR_{elec}	-0.0299	-0.0263	-0.0076	-0.0153
	(0.0216)	(0.0224)	(0.0956)	(0.0736)
	[-0.0614, -0.0098]	[-0.0549, -0.0131]	[-0.0408, 0.0037]	[-0.0510, 0.0041]
Controls	Yes	Yes	Yes	Yes
Applicant FE	Yes	No	Yes	No
Other FEs	Yes	Yes	Yes	Yes
Technology FE	Yes	Yes	Yes	Yes
Country FE	Yes	Yes	Yes	Yes
Year FE	Yes	Yes	Yes	Yes
P-value Bootstrap-c PMR_{elec}	0.0076	0.0008	0.0672	0.0134
F-stat first stage	33.01	35.62	33.03	35.67
First Stage Bootstrap p-value	0.0425	0.0176	0.0426	0.0175
Observations	49,583	55,642	49,542	55,600

A.4 2SLS using regulation in Air Transport as instrument

Notes: 2SLS regressions. I report wild bootstrap cluster p-values in parentheses and wild bootstrap cluster 95% confidence intervals in square brackets, generated using boottest command in Stata (Roodman et al., 2019) for standard errors clustered at the country level (18 clusters). The estimated model is the same model presented in Section 4. The significance of the coefficient associated with PMR_{elec} is tested also relying on unstudentized wild bootstrapping (P-value Bootstrap-c PMR_{elec}). Note that when using clustered standard errors based on asymptotic theory the instrument appears to be strong enough, but the p-value of the first-stage regression obtained using wild bootstrap would lead us to conclude the opposite. This being the case, these estimates could suffer from weak-instruments issues.

A.5 Patent Family correction based on the lowest level of the indicator of interest

	(1)	(2)	(3)	(4)
Panel A: OLS estimates	Rad	Rad	Ori	Ori
Lag PMR_{elec}	-0.0190	-0.0192	-0.0029	-0.0068
	(0.0001)	(0.0000)	(0.0610)	(0.0568)
	[-0.0240, -0.0157]	[-0.0263, -0.0141]	[-0.0083, 0.0003]	[-0.0115, 0.0004]
Controls	Yes	Yes	Yes	Yes
Applicant FE	Yes	No	Yes	No
Technology FE	Yes	Yes	Yes	Yes
Country FE	Yes	Yes	Yes	Yes
Year FE	Yes	Yes	Yes	Yes
Observations	49,583	55,642	49,578	$55,\!636$
Panel B: 2SLS estimates (telecom)	Rad	Rad	Ori	Ori
Lag PMR_{elec}	-0.0191	-0.0147	-0.0012	-0.0038
	(0.0016)	(0.0114)	(0.6669)	(0.3810)
	[-0.0326, -0.0099]	[-0.0205, -0.0066]	[-0.0153, 0.0029]	[-0.0139, 0.0104]
Controls	Yes	Yes	Yes	Yes
Applicant FE	Yes	No	Yes	No
Other FEs	Yes	Yes	Yes	Yes
P-value Bootstrap-c PMR_{elec}	0.0002	0.0032	0.6679	0.3764
F-stat first stage	54.04	69.15	52.80	69.43
First Stage Bootstrap p-value	0.0023	0.0016	0.0023	0.0016
Observations	49,583	55,642	49,578	$55,\!636$

Notes: OLS regressions (Panel A) and 2SLS regression (Panel B). The table presents the results obtained selecting from each family only the patent associated to the lowest value of the indicator of interest. In Panel B regulation in telecommunication is used as instrument for regulation in the electricity sector. I report wild bootstrap cluster p-values in parentheses and wild bootstrap cluster 95% confidence intervals in square brackets, generated using boottest command (Roodman et al., 2019) for standard errors clustered at the country level (18 clusters). In Panel B, the significance of the coefficient associated with PMR_{elec} is tested also relying on unstudentized wild bootstrapping (P-value Bootstrap- PMR_{elec})

	(1)	(2)	(3)	(4)
Panel A: OLS regression	Rad	Rad	Ori	Ori
Lag PMR_{elec}	-0.0188	-0.0195	-0.0039	-0.0079
	(0.0009)	(0.0000)	(0.0443)	(0.0495)
	[-0.0245, -0.0140]	[-0.0255, -0.0148]	[-0.0100, -0.0003]	[-0.0128, -0.0001]
Controls	Yes	Yes	Yes	Yes
Applicant FE	Yes	No	Yes	No
Technology FE	Yes	Yes	Yes	Yes
Country FE	Yes	Yes	Yes	Yes
Year FE	Yes	Yes	Yes	Yes
Observations	51,216	57,368	50,741	56,875
Panel B: 2SLS regression	Rad	Rad	Ori	Ori
Lag PMR_{elec}	-0.0182	-0.0133	-0.0028	-0.0040
	(0.0049)	(0.0381)	(0.2480)	(0.4915)
	[-0.0309, -0.0070]	[-0.0203, -0.0012]	[-0.0145, 0.0029]	[-0.0158, 0.0130]
Controls	Yes	Yes	Yes	Yes
Applicant FE	Yes	No	Yes	No
Other FEs	Yes	Yes	Yes	Yes
P-value Bootstrap-c PMR_{elec}	0.0098	0.0292	0.2798	0.4606
F-stat first stage	55.95	70.87	54.97	69.31
First Stage Bootstrap p-value	0.0022	0.0016	0.0023	0.0016
Observations	51,216	57,368	50,741	56,875

A.6 Including patents with inventors in different countries

Notes: OLS regressions (Panel A) and 2SLS regression (Panel B). The table presents the results when shared applications with inventors in different countries are included in the analysis. To include these applications I allocate them to the country where the market is less regulated among inventors' countries. The rationale behind this decision is that, given the research hypothesis of the paper, it is interesting to see if the highest level of liberalisation to which the invention is exposed has an effect on the search space of the patent. The estimated model is the same presented in Section 4. In Panel B regulation in telecommunication is used as instrument for regulation in the electricity sector. I report wild bootstrap cluster p-values in parentheses and wild bootstrap cluster 95% confidence intervals in square brackets, generated using boottest command (Roodman et al., 2019) for standard errors clustered at the country level (18 clusters). In Panel B, the significance of the coefficient associated with PMR_{elec} is tested also relying on unstudentized wild bootstrapping (P-value Bootstrap-c PMR_{elec})

FONDAZIONE ENI ENRICO MATTEI WORKING PAPER SERIES

"NOTE DI LAVORO"

Our Working Papers are available on the Internet at the following address: <u>https://www.feem.it/pubblicazioni/feem-working-papers/</u>

"NOTE DI LAVORO" PUBLISHED IN 2022

- 1. 2022, Daniele Crotti, Elena Maggi, Evangelia Pantelaki, <u>Urban cycling tourism. How can bikes and</u> public transport ride together for sustainability?
- 2. 2022, Antonio Acconcia, Sergio Beraldo, Carlo Capuano, Marco Stimolo, <u>Public subsidies and</u> cooperation in research and development. Evidence from the lab
- 3. 2022, Jia Meng, ZhongXiang Zhang, <u>Corporate Environmental Information Disclosure and Investor</u> <u>Response: Empirical Evidence from China's Capital Market</u>
- 4. 2022, Mariagrazia D'Angeli, Giovanni Marin, Elena Paglialunga, <u>Climate Change, Armed Conflicts and</u> <u>Resilience</u>
- 5. 2022, Davide Antonioli, Claudia Ghisetti, Massimiliano Mazzanti, Francesco Nicolli, <u>The economic</u> returns of circular economy practices
- 6. 2022, Massimiliano Mazzanti, Francesco Nicolli, Stefano Pareglio, Marco Quatrosi, <u>Adoption of Eco</u> <u>and Circular Economy-Innovation in Italy: exploring different firm profiles</u>
- 7. 2022, Davide Antonioli, Claudia Ghisetti, Stefano Pareglio, Marco Quatrosi, <u>Innovation, Circular</u> <u>economy practices and organisational settings: empirical evidence from Italy</u>
- 8. 2022, Ilenia Romani, Marzio Galeotti, Alessandro Lanza, <u>Besides promising economic growth, will the</u> <u>Italian NRRP also produce fewer emissions?</u>
- 9. 2022, Emanuele Ciola, Enrico Turco, Andrea Gurgone, Davide Bazzana, Sergio Vergalli, Francesco Menoncin, Charging the macroeconomy with an energy sector: an agent-based model
- 10. 2022, Emanuele Millemaci, Alessandra Patti, Nemo propheta in Patria: Empirical Evidence from Italy
- 11. 2022, Daniele Valenti, Andrea Bastianin, Matteo Manera, <u>A weekly structural VAR model of the US</u> <u>crude oil market</u>
- 12. 2022, Banchongsan Charoensook, <u>On Efficiency and Stability in Two-way Flow Network with Small</u> Decay: A note
- 13. 2022, Shu Guo, ZhongXiang Zhang, <u>Green credit policy and total factor productivity: Evidence from</u> <u>Chinese listed companies</u>
- 14. 2022, Filippo Bontadini, Francesco Vona, <u>Anatomy of Green Specialisation: Evidence from EU</u> <u>Production Data, 1995-2015</u>
- 15. 2022, Mattia Guerini, Fabio Vanni, Mauro Napoletano, <u>*E pluribus, quaedam*</u>. Gross domestic product out of a dashboard of indicators
- 16. 2022, Cinzia Bonaldo, Fulvio Fontini, Michele Moretto, <u>The Energy Transition and the Value of</u> <u>Capacity Remuneration Mechanisms</u>
- 17. 2022, Giovanni Marin, Francesco Vona, <u>Finance and the Reallocation of Scientific, Engineering and</u> <u>Mathematical Talent</u>
- 18. 2022, Anna Laura Baraldi, Erasmo Papagni, Marco Stimolo, <u>Neutralizing the Tentacles of Organized</u> <u>Crime. Assessment of Anti-Crime Measure in Fighting Mafia Violence</u>
- 19. 2022, Alexander Golub, Jon Anda, Anil Markandya, Michael Brody, Aldin Celovic, Angele Kedaitiene, Climate alpha and the global capital market
- 20. 2022, Jlenia Di Noia, Agent-Based Models for Climate Change Adaptation in Coastal Zones. A Review
- 21. 2022, Alberto Gabino Martínez-Hernández, <u>System Dynamics modelling and Climate Change</u> <u>Adaptation in Coastal Areas: A literature review</u>
- 22. 2022, Pietro De Ponti, Matteo Romagnoli, <u>Financial Implications of the EU Emission Trading System: an</u> <u>analysis of wavelet coherence and volatility spillovers</u>
- 23. 2022, Laura Cavalli, Mia Alibegovic, Davide Vaccari, Andrea Spasiano, Fernando Nardi, <u>Sustainability</u> assessment of the public interventions supported by the ReSTART project in the CITI4GREEN framework
- 24. 2022, Filippo Maria D'Arcangelo, Giulia Pavan, Sara Calligaris, <u>The Impact of the European Carbon</u> Market on Firm Productivity: Evidence from Italian Manufacturing Firms.
- 25. 2022, Enrico Turco, Davide Bazzana, Massimiliano Rizzati, Emanuele Ciola, Sergio Vergalli, <u>Energy prices</u> shocks and macroeconomic stabilization policies in the MATRIX model
- 26. 2022, Mario Gilli, Filippo Giorgini, <u>Citizens Protests: causes and consequences. A Research on Regime</u> <u>Change and Revolutionary Entrepreneurs</u>

"NOTE DI LAVORO" PUBLISHED IN 2022

27. 2022, Filippo Maria D'Arcangelo, Marzio Galeotti, <u>Environmental Policy and Investment Location: The</u> Risk of Carbon Leakage in the EU ETS

28. 2022, Italo Colantone, Livio Di Lonardo, Yotam Margalit, Marco Percoco, <u>The Political Consequences of</u> Green Policies: Evidence from Italy

29. 2022, Filippo Belloc, Edilio Valentini, <u>Digging into the Technological Dimension of Environmental</u> <u>Productivity</u>

30. 2022, ZhongXiang Zhang, <u>China's Carbon Market: Development, Evaluation, Coordination of Local and</u> National Carbon Markets and Common Prosperity

31. 2022, Aurélien Saussay, Misato Sato, Francesco Vona, Layla O'Kane, <u>Who's fit for the low-carbon</u> transition? <u>Emerging skills and wage gaps in job and data</u>
32. 2022, Francesco Nicolli, Marianna Gilli, Francesco Vona, <u>Inequality and Climate Change: Two Problems</u>, OneSolution?

33. 2022, Davide Consoli, Giovanni Marin, Francesco Rentocchini, Francesco Vona, <u>Routinization, Within-</u>OccupationTaskChangesandLong-RunEmploymentDynamics

34. 2022, Chiara Lodi, Giovanni Marin, Marco Modica, <u>Fiscal policy response of local governments to</u> floods in Italy

35. 2022, C. Castelli, M. Castellini, E. Ciola, C. Gusperti, I. G. Romani, S. Vergalli, A review of macroeconomic models for the WEFE nexus assessment

 2022, Laura Bonacorsi, Vittoria Cerasi, Paola Galfrascoli, Matteo Manera, <u>ESG Factors and Firms' Credit Risk</u>
 2022, D. Bergemann, M. Bertolini, M. Castellini, M. Moretto, S. Vergalli, <u>Renewable energy communities</u>, <u>digitalization and information</u>

38. 2022, F. Vaccari, Competition in Signaling

39. 2022, F. Vaccari, Efficient Communication in Organizations

40. 2022, F. Vaccari, Influential News and Policy-making

41. 2022, A. Albertazzi, M. Ploner, F. Vaccari, Welfare in Experimental News Market

42. 2022, G. Novelli, Energy Dependency and Long-Run Growth

43. 2022, G. Cazzola, Orienting Flood Risk Management to Disaster Risk Creation: lessons from the Water Framework Directive

44. 2022, M. Romagnoli, <u>Can electricity liberalisation foster the development of radical clean energy</u> technologies

Fondazione Eni Enrico Mattei Corso Magenta 63, Milano – Italia

Tel. +39 02 403 36934

E-mail: letter@feem.it www.feem.it

